What is the possible role of PSA doubling time (PSADT) and PSA velocity (PSAV) in the decision-making process to initiate salvage radiotherapy following radical prostatectomy in patients with prostate cancer?

Piotr Milecki^{1,4}, Andrzej Antczak², Piotr Martenka³, Zbigniew Kwias²

¹Radiotherapy Department, Greater Poland Cancer Center, Poznań, Poland ²Chair of Urology, Medical University, Poznań, Poland ³Radiotherapy Department, Greater Poland Cancer Center, Poznań, Poland ⁴Department of Electroradiology, Medical University, Poznań, Poland

KEY WORDS

prostate cancer D prognostic factors D PSA doubling time D PSA velocity D salvage radiotherapy

ABSTRACT

This article is an attempt to present a contemporary view on the role of the kinetics of PSA levels as defined by PSA doubling time (PSADT) and PSA velocity (PSAV) in the decision-making process to initiate salvage radiotherapy in patients with prostate cancer after radical prostatectomy (RP).

The dynamics of the rise of PSA levels may be an early endpoint parameter, preceding the diagnosis of distant metastasis or death due to prostate cancer based on a single PSA determination. Thus, it seems reasonable to include the kinetics of PSA levels, apart from single PSA determination, in the decision-making algorithm. In a group of patients after RP, PSADT might be an early endpoint that could replace cause-specific survival rate as a late endpoint. PSADT allows distinguishing subgroups of patients at high risk of distant metastases and death, which in turn may lead to a change in the further treatment strategy. Therefore, patients with short PSA doubling time should become a subgroup, in which hormonal therapy should be considered. To date, there is no unanimous consent to accept the criteria of assessment of the dynamics of PSA levels as determinants of treatment in case of recurrences following RP. However, a number of non-randomized clinical trials in patients after RP suggest it would be useful to include these parameters in the decision-making process. For instance, a relationship was found between increased PSA velocity (>2 ng/mL/year) before initiation of oncological treatment and increased (12-fold) risk of death. A number of well-documented retrospective analyses show that PSADT is one of the most important parameters to describe the disease aggressiveness. It has to be stressed that single determination of PSA levels is much less precise in terms of describing the biological aggressiveness of prostate cancer than PSADT. Of course, the question regarding the need to include the PSA levels kinetic parameters as crucial elements of patient management algorithms can be answered in a definitive manner only by randomized clinical trials.

INTRODUCTION

One possible option for radical prostate cancer treatment is the highly efficacious radical prostatectomy (RP). However, treatment failures still occur in a significant percentage of patients [1, 2]. Therefore, methods to improve the treatment results continue to be sought; one of these methods is adjuvant radiotherapy (RT) [3, 4]. The initiation of RT in the prostate cancer treatment is based on the assessment of a number of prognostic factors, such as pT, pN, or post-operative margin status, all included in the histopathology protocol following the RP, and in addition on the determination of PSA levels [5]. In the group of patients after RP with high risk of prostate cancer progression, there is a possibility to initiate early RT without signs of biochemical progression, delayed RT initiated upon detection of biochemical progression, or hormonal therapy in case of systemic progression [6, 7]. In case when recurrence risk factors are present (positive post-operative margin, infiltration outside the prostatic capsule – pT3a, infiltration of seminal vesicles – pT3b), the preferred treatment method is early RT [8, 9]. The use of salvage RT, reserved for cases of biochemical progression, is a less efficacious method [10, 11]. However, one must assume that the number of patients, in whom early RT was abandoned despite the presence of the disease progression risk factors defined in the pathology report, could be significant. At the same time, it must be stressed that this group is highly heterogeneous as regards the biological aggressiveness of the neoplastic process. One of the subgroups of these patients consists of patients in whom tumor microdissemination undetectable by available diagnostic methods - had occurred before the surgery. Another subgroup of patients after RP may consist of patients at very high risk of systemic progression and, at the same time, at very low risk of local progression. And the last group, include patients with biochemical progression after RP with very high risk of local and at the same time very low risk of systemic progression. Therefore, RT in this subgroup (high risk of cell presence in the surgery site) is likely to be associated with therapeutic benefit. However, in clinical practice, it is very difficult to assign a patient to one of these groups, and therefore, additional tools, which would allow to do it in the best possible way, are being sought.

A patient after radical prostatectomy: diagnostic dilemmas

The goal of the surgical treatment is to remove the entire pool of tumor cells present in the prostate gland, in the seminal vesicles,

and - less commonly - in lymph nodes. However, local efficacy of RP is not always sufficient, leading to biochemical recurrence, preceding or accompanying a simultaneous local recurrence. Postoperative assessment of PSA levels is an early measure of RP efficacy. commonly used in clinical practice [12]. Unfortunately, single determination of PSA levels does not allow defining the failure site (two sequential determinations are used to define the biochemical failure). It must also be stressed that non-lesioned fragments of prostate gland may be retained after the surgery in a group of patients. This may result in the maintenance of PSA level above the accepted cut-off levels, which, when exceeded, suggest biochemical failure. In general, documentation of biochemical failure extorts definition of failure "geography". Firstly, we should define whether the neoplastic process is limited only to post-operative site, or to the site with the accompanying distant metastasis currently beyond the detection capacities of diagnostic methods, or whether it is only a "chip" of non-lesioned prostate? This list shows that basing the therapeutic decision on the pathology report and single determination of post-operative PSA levels suggestive of biochemical failure is still associated with high risk of initiating a suboptimal treatment.

In patients after RP, RT is an established adjuvant treatment method, leading to reduction of biochemical failure incidence by ca. 50% [13]. Unfortunately, most randomized clinical trials conducted to date did not bring evidence of a statistically significant effect of early RT on the improvement in total survival rates [14-18]. An exception is the analysis of survival rates of patients in the SWOG study, presented at the ASCO conference in 2008 [19]. Despite an overwhelming number of publications suggesting that early RT is more efficacious than salvage RT, many urology centers hold to the belief that since there are no clinical trial results explicitly suggesting an increase in total survival rates thanks to early RT, it should be used only in case of biochemical progression. Unfortunately, diagnostic tools helping to define the source of biochemical failure, and thus helpful in gualifying patients for RT, are imprecise due to their low sensitivity and specificity. The role of the simplest of these tools, i.e. the digital rectal examination (DRE), in the diagnosis of local recurrence in patients after RP is very limited; in case of lack of biochemical recurrence this examination provides no useful information at al. [20]. Also, the usefulness of imaging examinations in defining local recurrences is low, even in cases of significant increase in PSA levels exceeding 0.2 ng/mL. Despite the diagnosed biochemical recurrence, the sensitivity and specificity of TRUS procedures and, in case of determining dissemination or isolated local recurrence, examinations such as bone scan, CT, or MRI with surface or anal coils are of limited importance [21, 22]. Hope for more sensitive and specific detection of micrometastases or minute local recurrences lies upon molecular imaging methods, namely [(11)C]choline PET/CT [23, 24] Recent reports support its use, showing good pathologic correlation with imaging data [25]. Distinguishing of failure source (isolated local recurrence versus distant metastasis \pm local recurrence) is important to the extent that it allows to initiate efficacious treatment in the form of salvage RT on one hand and, on the other hand, to avoid unnecessary RT in case of patients with a distant metastasis and to refer such patient to a clinical trial assessing novel systemic treatments.

One of the parameters most commonly used in clinical practice due to its availability and low acquisition costs is determination of the dynamics of PSA levels, expressed by PSA doubling time (PSADT) and PSA velocity (PSAV). Therefore, a number of articles were published in recent years with regard to the usefulness of these parameters in the therapeutic decision-making process in patients after RP. One of the most important research teams is the D'Amico team, which assessed the usefulness of measuring the kinetics of PSA levels prior to RP for the assessment of patients' fates after prostatectomy in a group of 1,095 prostate cancer patients [26]. The authors pointed that in 28% of patients, in whom the PSA velocity (PSAV) exceeded 2 ng/mL/year, a 10-fold increase in the risk of death due to prostate cancer than in the group of patients with PSAV <2 ng/mL/year. What's interesting, this risk was practically independent of other clinical and pathological parameters describing the prostate cancer. According to the authors, adjuvant RT in cases when PSAV exceeds 2 ng/mL/year brings little benefit due to large risk of disease dissemination. Therefore, authors think it would be advisable to consider initiation of systemic treatment in this group of patients (PSAV >2 ng/mL/year).

The next study by D'Amico et al. was very important in terms of defining the role of prognostic factors in patients after RP and RT [27]. In this study, clinical parameters associated with the risk of death due to prostate cancer in case of biochemical recurrence after radical treatment were singled out. With this purpose, an analysis of 8,669 patients (5,918 patients after RP, 2,751 patients after radical RP) was conducted, with mean observation time of 7.1 years after RT and 6.9 years after RP. The results of the statistical analysis showed that PSADT <3 months (found in 12% of patients in the operative treatment group and in 20% of patients in radiotherapy group) was an independent prognostic factor of the risk of death due to prostate cancer (HR = 19.6, Cl 95%; 12.5-30.9). Therefore, the authors claim that documentation of PSADT < 3 months indicates advisability of considering initiation of systemic treatment. In addition, the authors stress the potential usefulness of PSADT as an early endpoint, which might replace the assessment of prostate cancer-specific survival rates in clinical trials. The use of this parameter might lead to significant reduction in the waiting time with respect to the summaries of the results of clinical trials evaluating novel treatments. However, it must be highlighted that the most important premise stemming from this study is the fact that patients with short PSADT should constitute a group in which hormonal therapy or participation in clinical trials evaluating novel treatments should be considered. This is especially important since the probable cause of treatment failure is associated with micrometastases, present even before the radical treatment (RP).

Also the study by Zhou et al. assessed the usefulness of PSADT as a prognostic factor in patients after RP and radical RT. Based on observation of 1,159 patients with prostate cancer (498 patients after RP, 661 patients after radical RT), the PSA doubling time shorter than 3 months was associated with relative risk of death due to prostate cancer of 54.9 (16.7–180.0) in RP patients, and 12.8 (7.0–23.1) in radical RT patients [28].

Tollefson of Mayo Clinic analyzed the treatment results of 1,064 patients after RP. For analytic purposes, the author differentiated three disease progression risk subgroups: a high risk subgroup, when PSADT was shorter than 12 months; a medium risk subgroup, when PSADT was between 1 and 10 years; and low risk subgroup, when PSADT was longer than 10 years [29]. The relative risk of distant metastases was 21.7 (8.0-58.6) and 6.8 (2.3-19.8) in the high and low risk groups, respectively. The author suggests that patients from the high risk group should firstly be the potential candidates for initiation of systemic therapies, while patients at medium risk (PSADT between 12 and 120 months) should be qualified for adjuvant RT, and patients at low risk remain under observation.

Freedland et al. analyzed the relationship between PSADT and the risk of death due to prostate cancer in a group of 5,096 patients after RP [3]. The statistical analysis performed by the authors allowed differentiating patients with PSADT of less than 3 months, in whom the relative risk of death due to prostate cancer was 27.48 (10.66-70.85). In the subgroup of patients with PSADT ranging from 3.0 to 8.9 months, the risk of death was 8.76 (3.74-20.50), while in the subgroup of patients with PSADT between 9.0 and 14.9 months, the risk of death was 2.44 (0.88-6.81). Thus, a question arises, whether early hormonal therapy may improve the survival of patients in the high failure risk subgroup? Experience gathered to date suggest a potential possibility of such an effect, but the lack of results of randomized clinical trials assessing this aspect of hormonal therapy does not allow routine recommendation of this treatment in clinical practice. Thus, probably future RCTs would evaluate all aspects of early HT in patients at high risk of prostate cancer progression determined on the basis of the PSADT.

PSADT and PSAV seem to be attractive parameters that might significantly improve the optimization of treatment selection process in patients after RP. However, it must be kept in mind that calculation of these parameters requires observation of patients in whom biochemical progression was observed. On the other hand, it is a commonly held belief that early initiation of salvage RT, i.e. at the lowest possible PSA levels, is most efficient. Thus, based on the available clinical data it is impossible to assess how the length of the waiting period required for PSA measurements for PSADT or PSAV calculation might negatively affect the results of salvage RT. It is possible that in the future, the use of determination of PSA dynamics based on PSA determinations in the range of 0 ng/mL to 0.2 ng/mL would allow early determination of the "geography" of the biochemical failure.

CONCLUSIONS

To sum up, it must be stated that PSADT is a very useful tool for defining subpopulations of patients after RP in case of biochemical failure. This endpoint may be used as a potential tool for proposing local or systemic therapy in a subgroup of patients at high risk of distant metastases. Such differentiation would allow abandoning systemic therapy in patients at low risk of systemic progression while proposing salvage RT as highly efficacious treatment method.

Based on the review of studies assessing the kinetics of PSA level changes, it can be stated that in patients with the presence of biochemical recurrence, PSADT may be an early treatment efficacy endpoint, which might potentially replace the assessment of cause-specific survival rates, especially in the clinical trials [27, 29, 30]. The most important premise stemming from this bibliographical review is that patients with short PSADT should constitute a group in which hormonal therapy should be considered, while patients with long PSADT should be destined for salvage RT.

REFERENCE

- Han M, Partin AW, Pound CR, et al: Long-term biochemical disease-free and cancer-specific survival following anatomic radical retropubic prostatectomy: the 15-year Johns Hopkins experience. Urol Clin North Am 2001; 28: 555-565.
- Bianco FJ Jr, Scardino PT, Eastham JA: Radical prostatectomy: long-term cancer control and recovery of sexual and urinary function ('trifecta'). Urology. 2005; 66 (Suppl. 5): 83-94.
- Valicenti RK, Gomella LG, Perez CA: Radiation therapy after radical prostatectomy: a review of the issues and options. Semin Radiat Oncol 2003; 13 (2): 130-40.
- Wiegel T, Bottke D, Steiner U et al: Phase III postoperative adjuvant radiotherapy after radical prostatectomy compared with radical prostatectomy alone in pT3 prostate cancer with postoperative undetectable prostate-specific antigen: ARO 96-02/AUO AP 09/95. J Clin Oncol. 2009; 27: 2924-2930.

- Freeland SJ, Humphreys EB, Mangold LA et al: *Risk of prostate cancer-specific mortality following biochemical recurrence after radical prostatec-tomy*. JAMA 2005; 294: 433-439.
- Jereczek-Fossa BA, Orecchia R: Evidence-based radiation oncology: Definitive, adjuvant and salvage radiotherapy for non-metastatic prostate cancer. Radiother Oncol 2007; 84 (2): 197-215.
- 7. Boorjian SA, Karnes RJ, Crispen PL et al: *Radiation therapy after radical prostatectomy: impact on metastasis and survival.* J Urol 2009; 182: 2708-2714.
- 8. Aus G, Abbou CC, Bolla M et al: *EAU Guidelines on Prostate Cancer*. Eur Urol 2005; 48: 546-551.
- National Comprehensive Cancer Network, version 1.2011, 12/13/10. Visited online 14.04.2011.
- Stephenson AJ, Shariat SF, Zelefsky MJ et al: Salvage radiotherapy for recurrent prostate cancer after radical prostatectomy. JAMA 2004; 291 (11): 1325-1332.
- Cotter SE, Chen MH, Moul JW et al: Salvage radiation in men after prostatespecific antigen failure and the risk of death. Cancer 2011 Mar 22; doi: 10.1002/cncr.25993. [Epub ahead of print].
- 12. Sandler HM, Eisenberger MA: Assessing and treating patients with increasing prostate specific antigen following radical prostatectomy. J Urol 2007; 178: 20-24.
- Pasquier D, Ballereau C: Adjuvant and salvage radiotherapy after prostatectomy for prostate cancer: a literature review. Int J Radiat Oncol Biol Phys 2008 15; 72 (4): 972-979.
- Hagan M, Zlotecki R, Medina C et al: Comparison of adjuvant versus salvage radiotherapy policies for postprostatectomy radiotherapy. Int J Radiat Oncol Biol Phys 2004; 59: 329-340.
- Trock BJ, Han M, Freedland SJ, Humphreys EBet al: Prostate cancer-specific survival following salvage radiotherapy vs observation in men with biochemical recurrence after radical prostatectomy. JAMA 2008; 299: 2760-2769.
- Leibovich BC, Engen DE, Patterson DE et al: *Benefit of adjuvant radiation* therapy for localized prostate cancer with a positive surgical margin. J Urol 2000; 163: 1178-1182.
- Vargas C, Kestin LL, Weed DW et al: Improved biochemical outcome with adjuvant radiotherapy after radical prostatectomy for prostate cancer with poor pathological features. Int J Radiat Oncol Biol Phys 2005; 61: 714-724.
- Bolla M, van Poppel H, Collette L et al: *Postoperative radiotherapy after radical prostatectomy: a randomised controlled trial (EORTC trial 22911).* Lancet 2005; 366: 572-578.
- Thompson IM, Tangen CM, Paradelo J et al: Adjuvant Radiotherapy for Pathological T3NOMO Prostate Cancer Significantly Reduces Risk of Metastases and Improves Survival: Long-Term Follow up of a Randomized Clinical Trial. J Urol 2009; 181: 956-962.
- Pound CR, Partin AW, Eisenberger MA et al: Natural history of progression after PSA elevation following radical prostatectomy. JAMA 1999; 281: 1591-1597.
- Johnstone PA, Tarman GJ, Riffeburgh R et al: *Yield of imaging and scintig-raphy assessing biochemical failure in prostate cancer patient*. Urol Oncol 1997; 3: 108-113.
- Sella T, Schwartz UH, Swindle PW: Suspected local recurrence after radical prostatectomy endorectal coil MR imaging. Radiology 2004; 231: 379-385.
- Rinnab L, Simon J, Hautmann RE et al: [(11)C]choline PET/CT in prostate cancer patients with biochemical recurrence after radical prostatectomy. World J Urol 2009; 27: 619-625.
- 24. Vees H, Buchegger F, Albrecht S et al: *18F-choline and/or 11C-acetate positron emission tomography: detection of residual or progressive subclinical disease at very low prostate-specific antigen values (<1 ng/mL) after radical prostatectomy.* BJU Int 2007; 99: 1415-1420.
- 25. Schilling D, Schlemmer HP, Wagner PH, Bottcher P et al: *Histological verification of 11C-choline-positron emission/computed tomographypositive lymph nodes in patients with biochemical failure after treatment for localized prostate cancer.* BJU Int 2008; 102: 446-451.

- D'Amico AV, Chen MG, Roehl KA et al: Preoperative PSA Velocity and the risk of death from prostate cancer after radical prostatectomy. NEJM 2004; 351: 125-135.
- 27. D'Amico AV, Moul JW, Carrol PR et al: *Surrogate end- point for prostate cancer-specific mortality after radical prostatectomy or radiation oncology.* J Nat Cancer Inst 2003; 95: 1376-1383.
- Zhou P, Chen MH, McLeod D et al: Predictors of prostate cancer-specific mortality after radical prostatectomy or radiation oncology. J Clin Oncol 2005; 23: 6992-6998.
- 29. Tollefson MK, Szelak JM, Leibovich BC et al: *Stratification of patient risk based on prostate-specific antigen doubling time after radical retropubic prostatectomy.* Mayo Clin Proc 2007; 82: 422-427.
- Freedland SJ, Humphreys EB, Mangold LA: *Risk of prostate cancer-specific mortality following biochemical recurrence after radical prostatectomy*. JAMA 2005; 294: 433-439.

Correspondence

Piotr Milecki Radiotherapy Department Greater Poland Cancer Center 15, Garbary Street 61-866 Poznań, Poland phone: +48 61 885 05 00 piotr.milecki@wco.pl